Author’s response to reviews

Title: Paracentesis after central retinal artery occlusion: a tenable therapy?

Authors:

Achim Fieß (achim.fieß@gmx.de)
Ömer Cal (oemer.cal@hsk-wiesbaden.de)
Stephan Kehrein (stephan.kehrein@me.com)
Sven Halstenberg (sven.halstenberg@hsk-wiesbaden.de)
Inez Frisch (augenaerzte-ingelheim@frisch.de)
Ulrich H Steinhorst (augenaerzte-ingelheim@steinhorst.de)

Version: 2
Date: 8 December 2013

Author’s response to reviews:

Tables are included now and some revisions were done